Author's response to reviews

Title: Central modulation in cluster headache patients treated with occipital nerve stimulation: an FDG-PET study

Authors:

Delphine Magis (dmagis@chu.ulg.ac.be)
Marie-Aurélie Bruno (mabruno@alumni.ulg.ac.be)
Arnaud Fumal (Arnaud.Fumal@chu.ulg.ac.be)
Pierre-Yves Gerardy (pygy79@gmail.com)
Roland Hustinx (rhustinx@ulg.ac.be)
Steven Laureys (Steven.Laureys@ulg.ac.be)
Jean Schoenen (jschoenen@ulg.ac.be)

Version: 2 Date: 10 February 2011

Author's response to reviews:

Dear Editors,

This is the revised version of our article "Central modulation in cluster headache patients treated with occipital nerve stimulation: an FDG-PET study". As you suggested, the text was copyedited in order to improve the style of written English. We hope that our article now fits your language quality requirements.

With our best regards,
Delphine Magis and Jean Schoenen on behalf of the authors.